Author’s response to reviews

Title: The level of netrin-1 is decreased in newly diagnosed type 2 diabetes mellitus patients

Authors:

Chen Liu (670769638@qq.com)
Xian Ke (kxj_kxj17@163.com)
Ying Wang (1172068544@qq.com)
Xiu Feng (497085717@qq.com)
Qi Li (1506594326@qq.com)
Ying Zhang (1013327922@qq.com)
Jian Zhu (359474385@qq.com)
Qian Li (shygu@njmu.edu.cn)

Version: 4 Date: 21 May 2016

Author’s response to reviews:

Responds to the reviewer’s comments:

Reviewer 1:

1. Response to comment 1-8:

Response: Thanks for the referee’s kind suggestion. We have carefully reviewed our manuscript which possibly due to version problems of PDF leaded to some information loss in the process of uploading. We have made the modifications as your comments. In addition, as comment 2, we mean that "excluded" were patients in type 2 diabetes prescribed any oral hypoglycemic agents or insulin.
2. Response to comment: Please rewrite the fifth paragraph of the discussion section, lines 226-235.

Response: Thanks for the referee’s suggestion. We have rewritten the paragraph which was marked in the text according to reviewer’s suggestion.

3. Response to comment: The first sentence in the conclusion section should be omitted, since the study does not provide any evidence that netrin-1 is a predictive and prognostic marker of inflammation.

Response: Thanks for the referee’s suggestion. We have made change according to reviewer’s suggestion.

Reviewer 2:

Response to comment: The author should correct the kind of highlights into the text.

Response: Thanks for the referee’s suggestion. We upload two editions, one is the manuscript with highlights and the other is the modified edition.

We tried our best to improve the manuscript and made some changes. We appreciate for editors/reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comment and suggestions.